Author's response to reviews

Title: Leukocyte-derived microparticles and scanning electron microscopic structures in two fractions of fresh cerebrospinal fluid in amyotrophic lateral sclerosis; a case report

Authors:

Anne C Zachau (anne.zachau@karolinska.se)
Mikael Landén (Mikael.Landen@neuro.gu.se)
Fariborz Mobarrez (Fariborz.Mobarrez@ki.se)
Rolf Nybom (rolf.nybom@gmail.com)
Håkan Wallén (hakan.wallen@ds.se)
Lennart Wetterberg (lennart.wetterberg@ki.se)

Version: 6 Date: 27 June 2012

Author's response to reviews:

Reference #12 missing bold text - now added